Author's response to reviews

Title: Experiences in applying skills learned in a Mental Health First Aid training course: a qualitative study of participants' stories

Authors:

Anthony F Jorm (ajorm@unimelb.edu.au)
Betty A Kitchener (bettyk@unimelb.edu.au)
Stephen K Mugford (stephen.mugford@qgsr.com)

Version: 3 Date: 17 October 2005

Author's response to reviews:

We have made the following changes to the paper as requested:
1. We have checked that Title and Abstract and Author list is the same as in the submission.
2. The boxes in the Results have been removed and italics used instead.
3. Tables 2 and 3 have had tabs removed, text alignment has been made consistent and borders are visible.